Author's response to reviews

Title: Microarray analysis identifies a set of CXCR3 and CCR2 ligand chemokines as early IFNB-responsive genes in peripheral blood lymphocytes: an implication for IFNB-related adverse effects in multiple sclerosis

Authors:

Jun-ichi Satoh (satoj@ncnp.go.jp)
Yusuke Nanri (nanriyu@cc.saga-u.ac.jp)
Hiroko Tabunoki (tabunoki@my-pharm.ac.jp)
Takashi Yamamura (yamamura@ncnp.go.jp)

Version: 3 Date: 13 May 2006

Author’s response to reviews: see over
May 13, 2006

Melissa Norton, MD  
Editor-in-Chief, BMC Neurology

Dear Prof Norton:

RE: MS: 5705286909804629 Final version “Microarray analysis identifies a set of CXCR3 and CCR2 ligand chemokines as early IFNβ-responsive genes in peripheral blood lymphocytes in vitro: an implication for IFNβ-related adverse effects in multiple sclerosis”

Following formatting changes have been made.

1. The Website address is deleted from the text but cited in the reference list [23].

2. Table 2, along with Table 3, is each made into one table that requires one page.

3. Supplementary materials renamed as additional files 1-7 are mentioned in the text, all of which are listed at the end of the manuscript, accompanied by the file format, the title of the data, and a short description of the data.

Sincerely yours,

Jun-ichi Satoh, M.D., Ph.D., the corresponding author  
Department of Bioinformatics  
Meiji Pharmaceutical University  
2-522-1 Noshio, Kiyose, Tokyo 204-8588, Japan.  
E-mail: satoj@my-pharm.ac.jp or satoj@ncnp.go.jp

JS: